



# Article Optimized Financial Planning: Integrating Individual and Cooperative Budgeting Models with LLM Recommendations

I. de Zarzà <sup>1,2,3,\*</sup>, J. de Curtò <sup>1,2,3</sup>, Gemma Roig <sup>1,4</sup>, and Carlos T. Calafate <sup>2</sup>

- <sup>1</sup> Informatik und Mathematik, GOETHE-University Frankfurt am Main,
- 60323 Frankfurt am Main, Germany; decurto@em.uni-frankfurt.de (J.d.C.); roig@cs.uni-frankfurt.de (G.R.) <sup>2</sup> Departamento de Informática de Sistemas y Computadores, Universitat Politècnica de València,
  - 46022 València, Spain; calafate@disca.upv.es
- <sup>3</sup> Estudis d'Informàtica, Multimèdia i Telecomunicació, Universitat Oberta de Catalunya, 08018 Barcelona, Spain
- <sup>4</sup> HESSIAN Center for AI (hessian.AI), 64293 Darmstadt, Germany
- \* Correspondence: dezarza@em.uni-frankfurt.de

Abstract: In today's complex economic environment, individuals and households alike grapple with the challenge of financial planning. This paper introduces novel methodologies for both individual and cooperative (household) financial budgeting. We firstly propose an optimization framework for individual budget allocation, aiming to maximize savings by efficiently distributing monthly income among various expense categories. We then extend this model to households, wherein the complexity of handling multiple incomes and shared expenses is addressed. The cooperative model prioritizes not only maximized savings but also the preferences and needs of each member, fostering a harmonious financial environment, whether they are short-term needs or long-term aspirations. A notable innovation in our approach is the integration of recommendations from a large language model (LLM). Given its vast training data and potent inferential capabilities, the LLM provides initial feasible solutions to our optimization problems, acting as a guiding beacon for individuals and households unfamiliar with the nuances of financial planning. Our preliminary results indicate that the LLM-recommended solutions result in budget plans that are both economically sound, meaning that they are consistent with established financial management principles and promote fiscal resilience and stability, and aligned with the financial goals and preferences of the concerned parties. This integration of AI-driven recommendations with econometric models, as an instantiation of an extended coevolutionary (EC) theory, paves the way for a new era in financial planning, making it more accessible and effective for a wider audience, as we propose an example of a new theory in economics where human behavior can be greatly influenced by AI agents.

Keywords: financial planning; household finance; LLMs; budgeting; extended coevolutionary

# 1. Introduction

Financial planning [1,2] is an indispensable part of modern life. Whether it is an individual planning their monthly expenses or a family juggling with combined finances, effective budgeting is crucial for achieving long-term financial security and prosperity. However, with the proliferation of expenses and the intricate nature of contemporary economic systems, the endeavor of financial planning often feels overwhelming to many. Traditional methods, while systematic, sometimes fall short in accommodating the dynamic nature of financial needs, and often lack the ability to personalize recommendations to cater to individual or household-specific priorities.

We are now entering the age of artificial intelligence (AI) [3], a transformative force that has already revolutionized numerous sectors, from healthcare to entertainment. The vast inferential and predictive capabilities of AI, especially models like the large language



Citation: de Zarzà, I.; de Curtò, J.; Roig, G.; Calafate, C.T.; Optimized Financial Planning: Integrating Individual and Cooperative Budgeting Models with LLM Recommendations. *AI* **2024**, *5*, 91–114. https://doi.org/10.3390/ ai5010006

Academic Editors: Albert Y.S. Lam and Yanhui Geng

Received: 18 November 2023 Revised: 7 December 2023 Accepted: 18 December 2023 Published: 25 December 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). model (LLM) [4], offer a unique advantage in the realm of financial planning. By processing vast amounts of financial data and patterns [5], such models can provide tailored recommendations that not only adhere to general financial best practices, which are widely accepted principles or guidelines in financial planning that are considered effective for managing personal finances, but are also congruent with specific user inputs.

In this paper, we set out to bridge the traditional econometric methods of financial planning with the power of AI-driven recommendations. By considering the LLM's suggestions as a foundation , we refine and optimize budget allocations for both individuals and households. Our approach strives to achieve a balance between mathematical optimization and real-world practicality, ensuring that the solutions are not just economically sound but also feasible and user-friendly.

In practical terms, when artificial neural networks (ANNs), including DL and LLMs, are trained through supervised learning, they employ a learning rule that iteratively converges toward an equilibrium, finely balancing the dilemma of underfitting versus overfitting. Despite their ability to discern nonlinear continuous transformations, their operational mechanics do not inherently encapsulate an internal representation of the nonlinear relationships between training exemplars [6]. The consequence of this limitation is twofold: firstly, when ANNs process data with inter-relations confined within Euclidean boundaries, they excel; secondly, when confronted with data existing beyond these confines, the results may not only be suboptimal but also unpredictably aberrant. This limitation is similar, in many respects, to the constraints encountered in traditional statistical regression methods.

In the subsequent sections, we will delve into the methodologies that we have developed for individual and cooperative financial planning, elucidate on the integration of LLMs in these methods, and present the results of our preliminary tests. Through this exploration, we hope to usher in a new paradigm in financial planning, one that is more accessible, efficient, and in tune with the unique needs of today's diverse society. That being said, it is also necessary to recognize the potential limitations and inherent risks of AI, such as errors and hallucinations, and the strategies that can be implemented to mitigate these issues, such as rigorous validation processes, the constant monitoring of AI outputs, and the incorporation of fail-safes and human oversight.

The essence of this study lies in addressing a critical challenge: the democratization of financial planning in a landscape riddled with complexity and unpredictability. Our work is driven by the imperative to equip individuals and households with the tools to navigate the economic maze toward sustainable financial health. The problem statement is twofold—firstly, the need to simplify the intricacies of financial decision making for the non-expert and, secondly, to create an adaptive system that is resilient to the volatile nature of economic dynamics.

Our motivation is rooted in the recognition that financial security and literacy are cornerstones of individual autonomy and societal well-being. We propose an innovative solution that intersects the predictive acuity of LLMs with the foundational constructs of econometrics. The mathematical core of our approach is to optimize a utility function *U* that encapsulates the financial health of an entity, of the form:

$$U(I,E) = \alpha \log(S) - \beta \sum_{o=1}^{n} w_o \log(E_o)$$
<sup>(1)</sup>

where *I* is income, *S* is savings, *E* represents expenses across *n* categories,  $w_o$  are the personalized preference weights, and  $\alpha$ ,  $\beta$  are parameters calibrating the trade-off between savings and expenditures. Through this formulation, we seek to craft a system that not only provides bespoke financial advice but also evolves with the user's changing economic scenario, embodying the fluidity required for real-world financial planning.

The problem at the heart of our research is the incremental complexity of personal financial management in a rapidly evolving economic landscape. In an age where financial instruments and markets are increasingly intricate, the average individual is at a significant disadvantage when attempting to plan for financial security. This challenge is not merely

academic but a pressing real-world issue that affects the quality of life and long-term well-being of individuals and families. The need for accessible financial planning is further highlighted by the widening gap in financial literacy, which can lead to poor financial decisions and increased economic inequality.

Our work is motivated by the imperative to render financial planning not only more approachable but also more aligned with the diverse and dynamic financial goals of individuals. We aim to address this by leveraging the computational prowess of LLMs to distill complex financial data into actionable insights. The mathematical underpinning of our approach is an optimization framework, tailored to the nuances of personal finance, which can be represented as:

$$\max\{U(\mathbf{x}): \mathbf{x} \in \mathcal{F}, \ \mathcal{G}(\mathbf{x}) \le 0, \ \mathcal{H}(\mathbf{x}) = 0\}$$
(2)

Here, **x** symbolizes the financial decision variables, *U* is the utility function that quantifies financial health,  $\mathcal{F}$  defines the feasible set based on financial constraints,  $\mathcal{G}$  represents inequality constraints (such as budget limits), and  $\mathcal{H}$  captures equality constraints (such as income–expense balance). Our motivation is to implement a system that empowers users with personalized, intelligent recommendations, thus equipping them to make informed financial choices in pursuit of economic resilience and prosperity.

The remainder of this paper is structured as follows: Section 2 provides related work. Section 3 focuses on the materials and methodology adopted, presenting the foundational aspects and research methodology pertinent to the optimization models. In Section 4, we explore the significance of LLMs in individual budgeting, emphasizing their recommendationdriven approach to financial planning. Section 5 offers an in-depth look into cooperative financial planning for households, showcasing the synergistic view between LLM recommendations and household members' financial priorities. The subsequent Section 6 presents the intricacies of cooperative budgeting within an established extended coevolutionary (EC) framework, elucidating its innovative facets. Moving forward, Section 7 describes our simulation framework, detailing how LLM-informed decisions can be simulated and analyzed, while also highlighting in Section 7.3 the importance of long-term planning dependencies to further develop the simulated framework. Finally, in Section 8, we draw our conclusions, recapitulating the primary findings and looking ahead at promising directions for future work in this domain.

## 2. Related Work

The landscape of financial planning and optimization has been deeply enriched by the intersection of computational intelligence and economic theories. In this section, we contextualize our contributions within the spectrum of existing literature, delineating the evolution of financial planning methodologies from traditional models to AI-augmented frameworks.

Traditional financial planning has been heavily influenced by seminal works such as those of Markowitz [7] on portfolio optimization and Ando and Modigliani's life-cycle hypothesis [8]. The optimization of financial assets has been extensively studied, with methods ranging from linear programming [9] to more sophisticated stochastic models [10].

The infusion of behavioral economics into financial optimization models has provided a more thorough understanding of human financial behavior, as evidenced by the works of Thaler and others [11]. This has led to models that account for irrational decision making and heuristics used by individuals, adding complexity to the optimization problems [12].

The advent of AI [13,14] in financial planning has opened new avenues for personalized and adaptive financial advice. Early systems utilized rule-based expert systems [15], while recent advancements have leveraged machine learning for predictive analytics in finance [16].

The integration of LLMs [4] into financial planning is a relatively new and emerging field. Recent studies have begun to explore the capabilities of models like OpenAI's GPT

series and explainable AI techniques in generating financial narratives [17,18]. Our work extends this line of research by embedding LLM insights into the optimization framework, a novel approach that has not been extensively covered in the existing literature.

Comparative studies have highlighted the strengths and weaknesses of AI-driven financial tools compared to traditional methods [19–21]. However, there remains a gap in the literature regarding the efficacy of LLM recommendations within the cooperative budgeting context, an area our research directly addresses.

While existing works provide a foundational understanding of the financial optimization landscape, there remains a lacuna when it comes to the integration of real-time, AI-driven, personalized recommendations into these models. Our research fills this gap by not only integrating but also critically examining and refining the role of LLMs within financial planning and optimization frameworks.

In summary, our research builds upon the rich heritage of financial optimization while branching into the relatively uncharted territory of LLM-augmented financial planning. By doing so, we aim to contribute a significant advancement in the domain, tailored for the dynamic economic world of the 21st century.

### 3. Materials and Methodology

This research proposes an optimization model for individual and cooperative financial planning, both enhanced by the integration of large language model (LLM) recommendations. The two primary facets of this approach are: (1) foundational mathematical models that describe budgeting optimization problems, and (2) computational algorithms that incorporate the LLM's recommendations into financial decision-making processes.

The objective of maximizing savings by allocating monthly income across various expense categories is grounded in core principles of theory in household finance. This approach aligns with the life-cycle hypothesis of Ando and Modigliani (1963) [8], which posits that individuals plan their consumption and savings behavior over their lifetime to achieve financial stability. Furthermore, it reflects the principles of optimal portfolio selection from the theory of modern portfolio of Markowitz (1952) [7], adapted to the context of household budgeting, where the portfolio is a mix of expenses and savings.

Our model also incorporates the behavioral aspects of financial decision making as discussed by Xiao (2008) [22], acknowledging that individuals have varying preferences and subjective weights on different expense categories. Additionally, Browning and Lusardi's (1996) research on micro theories and micro facts of household saving [23] provides empirical support for the relevance of optimizing savings behavior at the household level.

For the cooperative financial recommendation model, we draw on these individual financial behaviors and extend them to a household context, considering the collective utility maximization and shared financial responsibilities, resonating with the cooperative game theory in household economics [24].

# Individual Financial Recommendation Model

Objective: Our primary goal is to find the best allocation of an individual's monthly income across various expense categories while maximizing their potential savings [7,8,22,23].

Variables:

- *I* : Monthly income.
- $E_z$ : Expense for category z (e.g., rent, groceries, utility bills, entertainment).
- *S* : Savings for the month.

Objective Function: This function emphasizes maximizing savings after covering all necessary expenses.

Maximize 
$$S = I - \sum_{z} E_{z}$$
. (3)

Constraints: The constraints ensure non-negative expenses, savings, and that total expenses do not surpass the monthly income.

E

$$z \ge 0 \,\forall z,$$
 (4)

$$S \ge 0,$$
 (5)

$$\sum E_z \le I. \tag{6}$$

# **Cooperative Financial Recommendation Model (Household)** Objective:

In a household setting, our aim is to optimize combined budget allocation for all members. We ensure that each member's preferences and necessities are accommodated, aiming for an overall maximization of household savings.

Variables:

- *I*<sub>o</sub> : Monthly income of member *o*.
- $E_{zo}$  : Expense for category *z* by member *o*.
- *S*<sub>o</sub> : Savings for the month by member *o*.
- *T<sub>S</sub>* : Total combined savings of the household.

#### **Objective Function:**

This function prioritizes the overall savings of the household, given each member's income and expenses.

Maximize 
$$T_S = \sum_o (I_o - \sum_z E_{zo}).$$
 (7)

Constraints:

The constraints are extensions of the individual model but factor in multiple members.

$$E_{zo} \ge 0 \,\forall z, o, \tag{8}$$

$$S_o \ge 0 \,\forall o,\tag{9}$$

$$\sum E_{zo} \le I_o \,\forall o,\tag{10}$$

$$T_S = \sum_o S_o. \tag{11}$$

Preferences/Weights:

To personalize the model, we introduce weights  $w_{zo}$  representing the importance of an expense category z for member o. This ensures that individual preferences are not lost in the collective model.

Maximize 
$$T_S = \sum_o (I_o - \sum_z w_{zo} E_{zo}).$$
 (12)

Collaborative Constraint:

Factoring in shared expenses, such as rent, can lead to substantial savings. Therefore, we introduce a combined expense constraint.

$$E_{\text{rent, combined}} \le E_{\text{rent,1}} + E_{\text{rent,2}}.$$
 (13)

## **Incorporating LLM Recommendations**

The true innovation of our approach lies in blending traditional optimization techniques with AI-driven insights [25,26]. The LLM's vast knowledge base and advanced language processing capabilities can provide nuanced, context-aware recommendations for individuals and households.

Given a recommendation  $R_z$  from the LLM for an expense category z, we can adjust our initial financial allocations.

$$E_z \approx R_z \,\forall z.$$
 (14)

With these LLM-generated insights serving as a foundation, the optimization model can be fine-tuned at a later stage to suit specific needs, balancing mathematical accuracy with real-world financial insights.

### 3.1. Prospective Validation Framework for AI-Driven LLM Financial Recommendations

This framework can be designed to uphold the integrity and applicability of AI-driven insights, ensuring that they are not only relevant but also conform to the highest standards of financial prudence.

A multi-tiered verification process that scrutinizes the LLM output for reliability and truthfulness can be proposed. This process would involve:

- Expert Review: Each LLM-generated recommendation is evaluated by financial experts to ensure that it is consistent with established economic principles and real-world conditions. Experts may modify LLM suggestions to correct any inaccuracies or to better align with conservative financial practices.
- Contextual Analysis: The model contextualizes LLM responses by examining them
  against current financial trends, regulatory guidelines, and economic indicators. This
  ensures that the advice is not only mathematically sound but also practical, taking
  into account the socio-economic environment of the users.
- Risk Assessment: Risk analysis is conducted to evaluate the potential harm that could arise from following the LLM's advice. This includes stress testing the recommendations against various economic scenarios to gauge their resilience and adaptability.
- Ethical Considerations: An ethical review is performed to ensure that the LLM's
  advice does not inadvertently lead to financial decisions that could be detrimental to
  the user's long-term well-being or that could perpetuate biases.

The objective function within the optimization framework would be then defined as follows:

Maximize 
$$U(S, \mathbf{E}, C) = f(S) - g(\mathbf{E}, C)$$
 (15)

where  $U(S, \mathbf{E}, C)$  is the utility function dependent on savings *S*, a vector of expenses **E**, and a context vector *C* that includes both economic conditions and user-specific circumstances. The function f(S) quantifies the utility derived from savings, emphasizing the goal of maximization, while  $g(\mathbf{E}, C)$  is a penalty function that increases when expenses are either economically unsound or potentially harmful.

To integrate LLM insights effectively, a constraint adaptation strategy can be employed where LLM recommendations adjust the boundaries of the feasible solution space rather than dictating fixed allocations. This would allows us to maintain a flexible yet controlled optimization environment where LLM inputs serve as informed guidelines rather than rigid directives. The optimization problem would then become:

Maximize 
$$U(S, \mathbf{E}, C)$$
 (16)

subject to  $\mathbf{E}_{\min} \le \mathbf{E} \le \mathbf{E}_{\text{LLM}}$ , (17)

$$S + \sum \mathbf{E} \le I, \tag{18}$$

$$S, \mathbf{E} \ge 0, \tag{19}$$

where  $\mathbf{E}_{min}$  and  $\mathbf{E}_{LLM}$  are vectors of minimum and LLM-recommended expenses, respectively.

Through this meticulous approach, we can ensure that the financial planning methodologies proposed in this study are not only innovative and AI-enhanced but are also reliable, ethically responsible, and practically applicable in a variety of financial contexts. This methodology, with the proposal of a thorough validation mechanism, sets a precedent for the responsible integration of AI in the sensitive realm of personal finance.

#### 3.2. RAG Approach with LLM Recommendations

Additionally, to address the potential for LLM hallucination, the methodology could integrate a retrieval augmented generation (RAG) approach into the recommendation generation process. This sophisticated mechanism can enhance the robustness of the LLM's outputs by anchoring its responses in verifiable data, thereby significantly reducing the likelihood of generating spurious or misleading financial advice.

The RAG process can operate by first querying a large-scale dataset of financial information, regulatory guidelines, and historical economic data. The LLM then synthesizes this retrieved information with its pre-trained knowledge to generate recommendations that are not only contextually rich but also data-grounded. Here is how the RAG methodology could be incorporated:

- Data Retrieval: Prior to recommendation generation, the LLM queries a curated dataset to retrieve relevant financial data and historical trends that mirror the user's query context.
- Augmented Generation: The model combines its own language-understanding capabilities with the retrieved data to generate a draft recommendation. This step ensures that each piece of advice is informed by empirical data, thereby enhancing its accuracy and relevance.
- Validation and Refinement: The draft recommendation is then passed through a validation filter, which includes a comparison against up-to-date financial models and a check for compliance with current economic conditions.
- Expert Oversight: The final stage can involve human oversight, where financial experts
  review the LLM-generated advice to ensure that it adheres to professional standards
  and reflects sound financial judgment.

The incorporation of RAG into our recommendation generation process can be mathematically represented in the optimization model as an additional set of constraints that the LLM's responses must satisfy:

$$RAG(\mathbf{E}, \mathbf{D}) \to \mathbf{E}_{\text{LLM}}$$
 (20)

where  $RAG(\mathbf{E}, \mathbf{D})$  represents the RAG process with expenses **E** and retrieved data **D**, mapping to the LLM's recommended expenses  $\mathbf{E}_{LLM}$ .

Incorporating the RAG methodology into the LLM framework could significantly mitigate the risk of hallucinations by ensuring that the model's recommendations are continuously grounded in empirical evidence. It can also provide an additional layer of confidence in the model's outputs, reinforcing their reliability for end-users seeking sound financial planning advice.

## 4. LLM-Informed Budgeting

In the modern era of data-driven solutions, using an LLM to provide financial advice offers an innovative approach to household budgeting. Given the intricate nature of individual spending habits and the myriad of ways in which funds can be allocated, using the power of AI can significantly enhance personalized financial recommendations.

The value addition of our research to the field of financial planning lies in the novel synthesis of econometric modeling and state-of-the-art AI. By integrating the insights from LLMs into traditional financial planning frameworks, we provide a multi-faceted approach that aligns with the contemporary demand for personalized financial advice. This integration heralds a significant shift from one-size-fits-all financial advisories to solutions that respect individual financial narratives and the unique economic contexts that they exist in.

Our models offer significant improvements over existing methodologies by incorporating behavioral economic factors and real-time data analysis, thereby delivering a more granular and responsive financial planning tool. The mathematical model that encapsulates our approach can be represented as:

$$\max_{\mathbf{x}} U(\mathbf{x}, \boldsymbol{\theta}) \quad \text{subject to} \quad \mathbf{x} \in \mathcal{F}(\boldsymbol{\Psi}), \ \mathcal{G}(\mathbf{x}, \boldsymbol{\Phi}) \le 0, \ \mathcal{H}(\mathbf{x}, \boldsymbol{\Lambda}) = 0$$
(21)

Here, **x** denotes the vector of decision variables for financial planning, *U* is the utility function enhanced by LLM insights  $\theta$ ,  $\mathcal{F}$  represents the feasible region influenced by external data  $\Psi$ , and  $\mathcal{G}$ ,  $\mathcal{H}$  are the sets of constraints parameterized by  $\Phi$  and  $\Lambda$ , respectively, signifying economic conditions and user preferences. The LLM's capability to digest and interpret large volumes of data transcends the capabilities of traditional methods, thus enabling our model to adaptively recalibrate recommendations in accordance with the fluctuating economic indicators and personal financial goals.

Moreover, the transformative potential of our methodology becomes particularly salient when considering the adaptability of LLMs to a changing information landscape, a feature that conventional financial planning tools lack. By continually learning from new data, our model remains at the forefront of financial planning innovation, marking a decisive step toward a more intelligent, adaptable, and personalized financial planning paradigm.

The approach leverages the capabilities of OpenAI's GPT-4 model to analyze a user's financial data and provide a bespoke recommendation based on that data. The process involves:

- 1. Data Input: Users provide a breakdown of their monthly income and primary expenses.
- 2. LLM Analysis: The model evaluates the data in the context of sound financial planning.
- 3. Recommendation Generation: The LLM then offers a detailed recommendation on how the user might best allocate their remaining funds.

The Python script, get\_financial\_recommendation, takes a string representation of the user's financial data and attempts to retrieve or generate a recommendation. Key features of the script include:

- Caching Mechanism: To enhance efficiency, the script maintains a recommendation\_cache. If the user's financial data match a previous entry, a stored recommendation is returned without invoking the LLM, thereby saving computational resources and time.
- Robust Error Handling: Recognizing the potential for API errors or timeouts, the script employs a retry mechanism with exponential backoff. This ensures that transient issues do not lead to immediate failures and offers a resilient approach.
- User-Friendly Output: The LLM's response is parsed to retrieve a clear, concise recommendation that is then returned to the user.

Here, we describe an example for clarity. Consider a user with a monthly income of USD 5000 and major expenses that include USD 1200 for rent, USD 400 for groceries, USD 200 for utilities, and USD 100 for entertainment. When they input these data, the LLM could provide a recommendation like:

"Based on your financial data for the month, after accounting for your major expenses, you have \$2100 remaining. I recommend allocating \$500 towards an emergency fund, \$300 towards investments, \$200 for miscellaneous expenses, and saving the remaining \$1100."

Advantages of the LLM Approach:

- 1. Personalization: Unlike generic financial advice, the LLM approach tailors recommendations to the individual's unique financial situation.
- 2. Scalability: The approach can cater to a vast number of users simultaneously, making it suitable for integration into financial planning apps or platforms.
- 3. Learning Over Time: As LLMs evolve and receive more training, the quality and depth of their recommendations can only improve.

While the current implementation provides robust financial recommendations, future versions could:

- Integrate real-time data sources to provide up-to-the-minute advice.
- Consider longer-term financial goals, such as buying a home or retirement planning, in its recommendations.
- Incorporate user feedback to continually refine and enhance recommendations.

In conclusion, using LLMs for household financial recommendations represents a significant step forward in the marriage of AI and financial planning. By offering personalized, scalable, and increasingly sophisticated advice, this approach promises to revolutionize the way individuals manage their finances.

Figure 1 delineates the proposed financial planning methodology that makes use of the capabilities of LLMs to enhance data analysis and supports the formulation of robust recommendations. This schematic presents a sequence of interconnected stages starting with the aggregation of financial data, proceeding to an in-depth analysis phase, which converge into the application of LLM insights for the generation of tailored financial recommendations. The iterative cycle is fundamental for maintaining the relevancy and efficacy of the financial plan, adapting dynamically to the evolving economic landscape and individual and cooperative financial objectives.



**Figure 1.** Flowchart illustrating the financial planning methodology with the integration of LLMs. The process starts with data collection, followed by analysis, integration of LLM insights, recommendation, implementation, and iterative review with monitoring to adjust the plan.

## 4.1. Verification Mechanism for LLM-Generated Financial Advice

In our approach, we could ensure the alignment of the LLM-generated recommendations with established financial truths by implementing a verification mechanism within the Python script: get\_financial\_recommendation . This mechanism can cross-reference the recommendations against a comprehensive financial database that includes a wide array of economic indicators, historical financial data, and regulatory compliance requirements. Here, we describe the process in detail:

- Recommendation Generation: The script first invokes the LLM to generate a preliminary financial recommendation based on the user's input data.
- Database Cross-Reference: Upon obtaining the LLM's output, the script automatically queries a financial database, seeking corroboration for the proposed financial strategies. This database is consistently updated to reflect the latest financial truths, including market trends, tax laws, and industry standards.
- Truth Alignment Check: The script compares the LLM's recommendation against the database entries to verify the accuracy and relevance of the advice. Recommendations that significantly diverge from the financial truths are flagged for further review.
- Feedback Loop for Adjustment: In cases where discrepancies arise, the script adjusts the recommendations by applying a set of heuristic rules derived from the financial truths. This iterative process ensures that the final advice provided to the user is both data-driven and aligned with financial realities.
- Human Expert Review: The last step involves a financial expert who reviews and, if necessary, refines the script's output to ensure full compliance with financial best practices.

Mathematically, this verification mechanism can be represented as a constraint satisfaction problem, where the generated recommendations R must satisfy a set of conditions C based on the truths T from the financial database D:

$$\forall r \in R, \exists t \in T : C(r, t) \rightarrow \text{Valid Recommendation.}$$
 (22)

Incorporating this truth-checking protocol could enhance the reliability of our LLMinformed budgeting system, ensuring that users receive recommendations that are not only personalized but also pragmatically sound and financially valid.

#### 4.2. Prompt Engineering for Improved LLM Financial Recommendations

To empower users in optimizing their interactions with the LLM, we can also introduce a guided prompt optimization feature within our LLM-informed budgeting system. This enhancement can educate users on constructing effective prompts, thereby improving the quality of the financial recommendations that they receive. Here is how we can incorporate this guidance into the user experience:

- Prompt Templates: We provide users with a collection of prompt templates that are designed based on successful financial inquiry patterns. These templates serve as starting points for users to articulate their financial data and questions.
- Prompt Feedback Mechanism: After users submit their prompts, our system analyzes the prompt structure and provides instant feedback on how to refine it for clearer communication and better-aligned financial recommendations.
- Adaptive Learning from User Interaction: As users interact with the system, it can
  employ ML algorithms to learn from the quality of recommendations generated from
  past prompts. The system then suggests modifications to the prompts for future
  queries, aiming to maximize the relevance and precision of the recommendations.
- Informative Tutorials and Tips: Alongside direct feedback, users have access to educational materials that explain the principles of effective prompt construction, including clarity, specificity, and the inclusion of relevant financial context.
- Expert System Insights: A built-in expert system analyzes user prompts and compares them with a database of optimized prompts and their outcomes. By highlighting key differences, the system offers actionable insights for users to improve their queries.

Mathematically, this process can be viewed as an optimization problem where the aim is to maximize the utility U of a prompt P with respect to the quality Q of the recommendation R as follows:

$$\max U(P) \quad \text{s.t.} \quad Q(R|P) \text{ is maximized.}$$
(23)

The function U(P) denotes the utility of the prompt, which is a function of its effectiveness in generating high-quality recommendations Q(R|P). By iteratively refining *P* based on system feedback and learning, we can ensure a continuous improvement in the prompt's utility.

#### 4.3. EC Theory for Household Finance: Beyond Euclidean Domains

The extended coevolutionary (EC) theory provides a sophisticated framework that allows us to navigate beyond the intrinsic limitations of ANNs. While these models might struggle with capturing the intricacies of non-Euclidean spaces due to their reliance on linear transformations within a Euclidean framework, EC theory embraces a more holistic view.

EC theory, through its foundational principles of coadaptation and coevolution, inherently accounts for the complex and dynamic inter-relationships characteristic of real-world scenarios, which often extend beyond linearly definable spaces. This adaptability is made possible by EC theory's capacity to model interactions within a multi-dimensional space that can effectively represent the high-level abstractions and nonlinear interdependencies of financial data.

$$\Delta \mathbf{x}(t+1) = F(\mathbf{x}(t), \mathbf{x}(t-1), \dots, \mathbf{x}(0)).$$
(24)

Here,  $\Delta \mathbf{x}(t + 1)$  represents the change in the strategy vector at time t + 1, and F encapsulates a nonlinear function of the strategy vectors at all previous time steps, capturing the essence of iterative learning and strategic adaptation over time. This function F is not restricted to the confines of a Euclidean space, enabling the EC model to explore a more diverse solution space and offering a robust mechanism for identifying equilibria in complex systems.

Furthermore, the EC theory facilitates the integration of external insights, such as those derived from LLMs, into its evolutionary process. By considering the LLM outputs as auxiliary data points, the EC model can refine its strategic outlook, effectively bridging the gap between the traditional optimization techniques and the high-dimensional space of AI-driven recommendations.

$$\mathbf{x}(t+1) = \mathbf{x}(t) + \gamma(\mathbf{R} - \mathbf{x}(t)) + \Delta \mathbf{x}(t+1).$$
(25)

In the equation above,  $\gamma$  represents the rate of integration of LLM recommendations denoted by **R** into the current strategy **x**(*t*). The term  $\Delta$ **x**(*t* + 1), as previously defined, ensures that the EC model dynamically evolves, reflecting both the insights from LLMs and the iterative learning process.

By leveraging the flexibility and multidimensional approach of EC theory, we can transcend the traditional limitations posed by ANNs. This progressive modeling approach not only aligns with the complex nature of financial ecosystems but also offers a robust platform for developing advanced financial planning tools that are both comprehensive and capable of evolving in tandem with the changing economic landscape.

#### 5. LLM Cooperative Budgeting and Financial Planning for Households

Cooperative budgeting emerges as an imperative when multiple individuals, such as partners or roommates, decide to combine their financial resources and manage expenses collectively. By jointly analyzing incomes and costs, households can better allocate resources, set shared financial goals, and plan for both short-term needs and longterm objectives.

The provided Python script, get\_cooperative\_financial\_recommendation, seeks to tap into the potential of LLMs to generate budgeting advice tailored for households that are pooling their resources.

Key Script Features:

- Caching Mechanism: Repetitive queries are bypassed by maintaining a recommendation\_cache. If data from a household match a prior request, the cached recommendation is returned, thereby conserving computational effort.
- Retry with Exponential Backoff: The script is robust against potential API timeouts
  or other exceptions. By employing a retry mechanism that increases the waiting time
  after each attempt, it provides resilience against transient issues.
- Tailored Recommendations: The response from the LLM is parsed to yield a succinct and actionable cooperative financial recommendation for the household.

## Sample Use Case

Consider a household comprising two members:

- Member 1 with an income of USD 5000, paying rent of USD 800 and spending USD 300 on groceries.
- Member 2 earning USD 4000, having a rent of USD 700 and grocery expenses amounting to USD 350.

Upon feeding these combined data to the LLM, a recommendation could emerge as:

"Given the combined income and individual expenses, the household has a remaining amount of \$7850 after rent and groceries. It is advised to allocate \$1500 for joint savings, \$1000 for utilities and shared amenities, \$500 towards entertainment and leisure, and set aside \$1000 as an emergency fund. The residual amount can be directed towards individual savings or other shared financial goals."

Benefits of Collaborative Budgeting:

- 1. Shared Responsibility: Distributing financial responsibilities can alleviate individual burdens and promote mutual accountability.
- 2. Strategic Decision Making: With pooled resources, households can make more informed choices about investments, large purchases, and savings.
- 3. Achievement of Common Goals: Be it buying a new home, planning a vacation, or preparing for family expansions, cooperative budgeting can expedite the realization of shared aspirations.

Enhancements for the Future:

The script, while effective, can be augmented further by:

- Incorporating feedback mechanisms where household members can adjust recommendations based on changing financial scenarios.
- Integrating data from financial institutions in real-time for more current and accurate recommendations.
- Introducing functionalities to handle more complex financial situations, such as investments, loans, and mortgages.

In essence, cooperative budgeting facilitated by AI-driven models can revolutionize the way households manage their finances, promoting a culture of transparency, mutual trust, and shared financial growth.

## 6. Cooperative Budgeting within the EC Framework

Cooperative budgeting for households represents a tangible instantiation of the extended coevolutionary (EC) framework's principles first introduced in [27]. By analyzing this in the context of the EC theory, we illuminate the synergistic confluence of gametheoretic models, LLMs, and adaptive learning in practical MAS scenarios.

Let us consider a household *H* consisting of *n* members, where each member *o* is an agent. Each agent has a financial profile characterized by its income  $I_o$  and expenses  $E_o$ . The collective budget of the household can be expressed as:

$$B_H = \sum_{o=1}^n I_o - E_o.$$
 (26)

The goal is to cooperatively allocate a common budget  $B_H$  across members to optimize some household-wide utility function  $U_H$ . This utility can be a measure of financial stability, savings, or any other metric significant to the household.

Incorporating the adaptive learning mechanism from the EC framework, each agent o iteratively adjusts its expenses  $E_o$  based on a derived utility  $U_o$  that takes into account the household's collective goal and individual preferences. The adjustment can be mathematically represented analogous to the previous equation as:

$$E_o(t+1) = E_o(t) + \alpha \nabla U_o(E_o(t), E_{-o}(t)).$$
(27)

Here,  $\alpha$  remains the learning rate, and  $\nabla U_o$  signifies the gradient of utility with respect to agent o's expenses, considering other agents' expenses as  $E_{-o}$ .

The strategy of agent o, specifically its expense  $E_o$ , can be influenced by recommendations from an LLM. Representing the LLM's recommendation for agent o at time t as  $R_{o,t}$ , and similar to the equation presented before, the new expense can be modeled as:

$$E_o(t+1) = (1-\beta)(E_o(t) + \alpha \nabla U_o(E_o(t), E_{-o}(t))) + \beta R_{o,t}.$$
(28)

Here,  $\beta$  designates the influence rate of the LLM on the agent's expense strategy.

Given that LLM recommendations, like budget suggestions, come with inherent uncertainties, a confidence measure  $C_{o,t}$  associated with each recommendation  $R_{o,t}$  should be considered. This is analogous to the previous equation, and the updated expense can be modeled as:

$$E_o(t+1) = (1 - \beta C_{o,t})(E_o(t) + \alpha \nabla U_o(E_o(t), E_{-o}(t))) + \beta C_{o,t} R_{o,t}.$$
(29)

Each household member may have distinct financial priorities and constraints, echoing the EC theory's emphasis on heterogeneous agents. The utility for each agent  $U_o$  would thus encapsulate individual preferences, like risk aversion or future financial goals.

The cooperative budgeting scenario for households aptly demonstrates the applicability of the EC theory in real-world contexts. This intersection of theoretical constructs with tangible applications showcases the robustness and versatility of the EC framework. The use of mathematical representations deepens our understanding, offering a quantitative lens to perceive the dynamics of the household as a multi-agent system influenced by adaptive learning and LLMs.

## Linear Optimization Model for Financial Planning with LLM Constraints

Another instance of the same financial planning scenario is the following, where we propose to use the LLM recommendation as bounds of the optimization model. We construct a linear optimization model to distribute the monthly income *I* across various expense categories while maximizing potential savings. The model is formulated as follows:

Let *I* represent the monthly income of a household, and  $E_z$  represent the expenses in category *z*, where *z* can be rent, groceries, utility bills, or entertainment. Let *S* denote the savings for the month. The goal is to maximize *S* subject to the constraints imposed by the financial data and recommendations from the LLM.

The objective function and constraints are presented as follows:

Maximize 
$$S = I - \sum_{z} E_{z}$$
 (30)

$$E_z \ge 0 \,\forall z$$
,(Non-negativity of expenses)(31) $S \ge 0$ ,(Non-negativity of savings)(32) $\sum_z E_z \le I$ ,(Total expenses do not exceed income)(33)

Additional constraints are derived from the LLM recommendations, which suggest percentages of income to be allocated toward each category. Let  $p_z$  be the percentage of income recommended for category z, and  $E_z^{\text{rec}}$  be the recommended expenses calculated as  $p_z \cdot I$ . We can introduce bounds based on these recommendations to guide the optimization:

$$E_z \ge E_z^{\min}$$
, (Minimum spending) (34)  
 $E_z \le E_z^{rec}$ , (LLM-recommended spending) (35)

where  $E_z^{\min}$  is a lower bound to ensure that each category meets essential spending requirements. The upper bound  $E_z^{\text{rec}}$  is influenced by the LLM's advice and represents an ideal spending target as suggested by the model.

#### 7. Simulation Framework: LLM-Informed Decisions

To comprehend the LLM's effectiveness in financial advice, it is essential to assess its recommendations. Let us delve into its suggestions for individual and cooperative financial management:

### 7.1. User Financial Recommendation

### OpenAI model: gpt4-0613

response = openai.ChatCompletion.create( model="gpt-4-0613", messages=[ {"role": "system", "content": "You are an expert financial planner with extensive knowledge in household finance and budgeting. Provide a detailed recommendation on how to allocate funds based on the given data."}, {"role": "user", "content": f"Here's my financial data for the month: {income\_data\_string}. Can you provide a recommendation on how I should spend or save the remaining funds?"} ])

The instruction given to the model focused on its expertise as a financial planner. This query is indicative of how the LLM would process individual financial data and provide a comprehensive recommendation.

# LLM User Financial Recommendation

Of course, based on your provided income and expenditure, your total spending adds up to \$1900 out of \$5000. This leaves you with a remaining balance of \$3100. Let's break down and allocate your remaining balance as follows:

1. \*\*Emergency Fund\*\*: If you don't already have an emergency fund, aim to build one. This should consist of 3–6 months' worth of living expenses to cover unexpected costs, such as a medical emergency or car repair. If you don't have such a fund pot, then it's good to allocate around 20% (\$620) of the remaining money towards it.

2. \*\*Retirement Savings\*\*: If you're not currently contributing to a retirement fund, it's a good idea to start. A general rule is to allocate 15–20% of your income towards your retirement. In this case from the remaining sum, consider allocating 20% (\$620). 3. \*\*Debt Payments\*\*: If you have any outstanding debt like student loans, credit card debt, etc, it's crucial to allocate a portion of your income towards paying these off. For now, from this remaining balance you could allocate 15% (\$465).

4. \*\*Personal Savings/Investments\*\*: It's always a good idea to invest in your future, whether it means buying a house, starting your own business, or taking a vacation. You could aim to allocate 20% (\$620) for this purpose.

5. \*\*Personal Spending\*\*: Set some money aside for personal use, shopping, or miscellaneous spending. Around 10% (\$310) could be allocated for this purpose.
6. \*\*Health and Fitness\*\*: Consider this as an investment in your health. Allocate

about 5% (\$155). 7. \*\*Education/Personal Development\*\*: Investing in oneself is always a wise decision. 5% of the remaining money which would be around \$155 could be dedicated

for this purpose. 8. \*\*Charity\*\*: Lots of successful financial planners recommend allocating a small portion of your income towards giving to others. Around 5% (\$155) should be again a good start.

Here's how the final allocation looks like:

- Emergency Fund: \$620
- Retirement Savings: \$620
- Debt payments: \$465
- Personal Savings/Investments: \$620
- Personal Spending: \$310
- Health and Fitness: \$155
- Education/Personal Development: \$155
- Charity: \$155

Analysis: The recommendation begins by acknowledging the user's financial state, ensuring transparency in the LLM's calculations. The model then meticulously divides the

leftover funds into eight categories. These categories represent the fundamental areas that most individuals need to consider in their financial planning. The proportional allocation seems balanced and encourages both saving and investing while also emphasizing personal well-being and community giving.

This framework aligns well with traditional financial planning advice, indicating the model's efficacy in this domain. Moreover, the clarity and structure provided by the model can greatly assist users in understanding their financial standing and planning their future expenditures.

7.2. Cooperative Financial Recommendation

OpenAI model: gpt4-0613

response = openai.ChatCompletion.create( model="gpt-4-0613", messages=[ {"role": "system", "content": "You are an expert financial planner with expertise in cooperative budgeting for households. Provide a comprehensive recommendation based on the combined financial data."}, {"role": "user", "content": f"Here's our combined household financial data: {household\_data\_string}. Can you recommend how we should collaboratively budget our expenses?"} ])

In this inquiry, the model was tasked with providing a comprehensive recommendation based on combined financial data, embodying the complexities of cooperative budgeting for households.

Analysis: The model adopts a detailed approach to cooperative financial planning, starting with a clear breakdown of combined income and essential expenses like housing and groceries. The percentages used to allocate various expenses, such as housing taking up 25–30% of the income, align with widely accepted financial guidelines.

A particularly noteworthy aspect of this recommendation is the inclusion of an emergency fund, highlighting the LLM's capacity to foresee potential financial pitfalls and its alignment with prudent financial practices.

The cooperative recommendation, much like the individual one, is comprehensive, detailed, and aligns with standard financial planning guidelines. This reinforces the model's reliability in producing accurate and actionable financial advice.

### LLM Cooperative Financial Recommendation

Absolutely. Here's a comprehensive budgeting plan based on the combined financial data:

1. \*\*Income Analysis:\*\* Your combined total monthly income is \$5000 + \$4000 = \$9000. 2. \*\*Housing Expenses:\*\* Your combined rent is \$800 + \$700 = \$1500, which is 16.67% of your income. Considering that housing should generally be around 25–30% of your income, you're in a good position here.

3. \*\*Grocery Expenses:\*\* Your combined grocery cost is 300 + 3350 = 650 per month. This cost is approximately 7.2% of your total income which is below the usual 10–15% range. You are so far managing this expense effectively but there may be room for improved nutrition or desired treats within this budget.

4. \*\*Savings:\*\* A good rule of thumb would be to save at least 20% of your income. In your case, that would be \$1800 per month.

5. \*\*Other Expenses:\*\* After accounting for rent, groceries and savings, you have 4050 = 100 - 100 - 100 - 100 - 100 - 1000 -

- \*\*Utilities: 5–10% of income:\*\* This would be about \$450-\$900.
- \*\*Healthcare: 5–10% of income:\*\* This should be approximately \$450–\$900.
- \*\*Transportation: 10–15% of income:\*\* This should ideally be between \$900– \$1350.
- \*\*Recreational & Personal Expenses: 5–10% of income:\*\* This is typically \$450– \$900.
- \*\*Retirement Contribution: At least 10% of income.\*\* This would be \$900 in your case.

This distribution may vary based on your personal needs, lifestyle, debt repayments, and obligations. It would be wise to keep track of your expenses to ensure you're in-line with your set budget.

6. \*\*Emergency Fund:\*\* It is also important to ensure that you have an emergency fund in place. Ideally, this fund should cover 3–6 months' worth of living expenses, and given your expense rate, this should be somewhere between \$10,800–\$21,600. If you do not have an adequate emergency fund in place, then portion of your savings should be allocated to this fund until the target is met.

End of the day, the goal is to ensure the income is distributed in a way that covers all the needs, while also ensuring there's financial growth and stability. Regularly revisit your budget and adjust accordingly based on changes in income, expenses, and financial goals.

## 7.3. The Significance of Long-Term Models in Financial Planning

Financial planning [28,29], at its core, is not just about budgeting for the next month or year: it is about laying a foundation for the entire life-cycle [30,31]. Decisions made today will inevitably impact financial security, flexibility, and wealth accumulation in the distant future. With this understanding, there arises an imperative need for long-term models in the domain of financial planning.

Long-term models bring foresight to financial decision making. By analyzing longterm implications of current financial data, these models offer recommendations that are not just about immediate benefit but also cater to future financial health. The importance of long-term models becomes even more pronounced in situations that involve uncertainties like changing job markets, economic fluctuations, or personal life events.

Here is a glimpse into the key benefits and functionalities of long-term models in the financial domain:

 Consumption Smoothing: Long-term models assist in balancing consumption patterns over time, ensuring that individuals do not overspend during prosperous periods or face hardship during lean times. By forecasting future income streams and financial obligations, long-term models promote consistent living standards throughout an individual's life.

- Life-cycle Modeling: These models simulate various stages of an individual's life, from early career stages, potential family planning, and mid-life crises, to retirement. By doing so, they help in preparing for financial needs and challenges typical to each phase.
- Risk Management: LLMs highlight potential risks by analyzing economic trends, inflation rates, and personal financial data. They help to devise strategies to mitigate these risks, be it through investments, insurance, or savings.
- Investment Guidance: Based on long-term financial goals and market predictions, LLMs provide guidance on where and how much to invest. This includes suggestions on diversifying portfolios, adjusting asset allocations, and more.
- Future Projections: One of the standout features of LLMs is their ability to project financial futures based on current data and decisions. They offer a roadmap of where one's finances could be in 5, 10, or even 30 years from now.

# 7.4. Long-Term Cooperative Financial Recommendation

# OpenAI model: gpt4-0613

response = openai.ChatCompletion.create( model="gpt-4-0613", messages=[ {"role": "system", "content": "You are an expert financial planner with expertise in life-cycle models and cooperative budgeting for households. Provide a comprehensive recommendation based on the combined financial data that considers consumption smoothing over the entire life-cycle."}, {"role": "user", "content": f"Here's our combined household financial data, along with some future projections and life events: {household\_data\_string}. Can you recommend how we should collaboratively budget our expenses?"} ])

Here is an example usage with more comprehensive data taking into account long-term dependencies:

## LLM Long-Term Cooperative Financial Recommendation

Using the life-cycle model involves setting your saving and spending habits with a long-term vision. Consumption smoothing is a key part of this, aiming to allocate resources in a way that maintains a consistent standard of living throughout your life. Here are my recommendations based on both your income and projected life events:

1. \*\*Start an Emergency Fund\*\*: Save a portion of your combined income consistently until you have about three to six months' worth of living expenses. Considering your incomes and basic monthly expenses, which total \$2150, I'd suggest allocating at least \$500/month for this initially.

2. \*\*Create a Child Fund\*\*: Since you're planning to have a child in 3 years, creating a dedicated child fund should be a priority. It will help cover everything from pregnancy-related medical costs to future schooling expenses. Let's allocate \$300/month for this fund.

3. \*\*Retirement Saving\*\*: Member 2 is planning to retire in 20 years, so a substantial part of the income should go towards retirement savings. If not already started, I propose an immediate beginning with an initial tier of at least \$600 a month.

4. \*\*Income Adjustment\*\*: The potential pay cut for Member 1 should be planned for in advance. To facilitate consumption smoothing, consider increasing savings or reducing non-essential expenses to accommodate the projected 10% pay cut.

5. \*\*Housing Expenses\*\*: Your total monthly rent is \$1500. If you stay in the current rental situation for the long term, there could be an opportunity to buy a house, as long as it's financially reasonable.

6. \*\*Daily Living and Unexpected Expenses\*\*: Allocate the rest of the income for your daily living expenses and unexpected costs (car repairs, medical expenses, etc.). In summary, this is a rough recommendation on which categories can the budget be divided:

- \*\*Emergency Fund\*\*: \$500/month
- \*\*Child Fund\*\*: \$300/month
- \*\*Retirement Savings\*\*: \$600/month
- \*\*Housing\*\*: \$1,500/month
- \*\*Groceries\*\*: \$650/month
- \*\*Daily Living and Unexpected Costs\*\*: The balance of your incomes after removing the above expenses (approximately \$2450).

This plan will require review and adjustment on a regular basis, at least yearly, to accommodate changes in income or new financial needs/goals. Remember, the key to the life-cycle model is flexibility and long-term planning.

As seen from our simulation framework, when LLMs like gpt-4-0613 are given a long-term perspective, they provide advice that aligns with the core tenets of financial planning. Whether it is individual or cooperative budgeting, the model's recommendations reflect an understanding of the intricate dance between current spending, future needs, and long-term financial growth.

Analysis: The long-term cooperative financial recommendation exemplifies the model's capacity to engage with complex financial scenarios that span across an individual's life-time. Notably:

- Consumption Smoothing: The model emphasizes consumption smoothing by suggesting a consistent standard of living. Its advice on adjusting income to accommodate a projected pay cut is particularly salient, as it showcases the model's ability to suggest proactive measures against potential financial setbacks.
- Emergency and Specific Funds: The inclusion of both an emergency fund and a child fund showcases the model's comprehensive approach. It highlights an understanding

of immediate needs (like emergencies) and future events (like planning for a child), thus promoting financial preparedness.

• Retirement Planning: The model's advice to prioritize retirement savings for Member 2, who plans to retire in 20 years, underscores its alignment with a long-term vision. It demonstrates an acute awareness of the time-sensitive nature of retirement planning.

Overall, the model's long-term cooperative financial recommendation reflects a good understanding of the life-cycle model's principles. It meticulously integrates immediate financial needs with long-term goals, showcasing its potential to serve as a valuable tool in the domain of financial planning.

#### 7.5. Simulation of Financial Planning with LLM Recommendations

Going beyond, we present a simulation that integrates financial recommendations from an LLM into the financial planning process of synthetic household datasets. Our approach involves generating synthetic financial profiles for ten households, encompassing monthly income and expenses across four categories: rent, groceries, utility bills, and entertainment. The monthly incomes are normally distributed around a mean of USD 5000 with a standard deviation of USD 1000. Expenses in each category are also generated following a normal distribution with category-specific means and variances.

Once the synthetic dataset is established, we employ an LLM to provide financial planning advice for each household. The LLM is prompted to analyze the financial data and recommend allocations as percentages of the household's monthly income. The advice is structured to include essential expenditures, savings, and discretionary spending, following sound financial planning principles.

The LLM's textual recommendations are parsed using regular expressions to extract the suggested allocation percentages for each financial category. These extracted values are then translated into monetary amounts based on the respective household's income and stored alongside the original data.

Subsequently, we apply a linear programming optimization model to each household's financial data, aiming to maximize savings while adhering to realistic constraints. This model takes into account minimum spending requirements for each expense category, ensuring that the optimized plan remains practical and sustainable.

Finally, we visualize the simulation results, comparing the original savings, LLMrecommended savings, and optimized savings with bounds. This comparison highlights the potential impact of AI-driven financial advice when combined with traditional optimization techniques.

The instance of a financial planning scenario in Figure 2 not only showcases the feasibility of utilizing AI in personal financial planning but also serves as a benchmark for evaluating the LLM's performance against traditional optimization models. The findings suggest that LLMs can effectively contribute to financial decision making, offering insights that align with recognized financial best practices.

The simulation involves several stages:

- 1. Generation of synthetic financial data for each household, with random distributions for income and expenses.
- 2. Retrieval of financial recommendations from an LLM, which analyzes the synthetic data and provides allocation percentages for various expense categories.
- Parsing of the LLM output to translate percentage-based recommendations into monetary allocations.
- Application of a linear optimization model to determine an optimized savings strategy for each household, independent of LLM recommendations, using fixed minimum spending bounds for each expense category.
- 5. Visual comparison of the households' original savings, the savings as recommended by the LLM, and the savings post-optimization, highlighting the potential of LLM guidance in enhancing financial decision making.

## LLM Recommendation

LLM\_output = Based on your income and expense data, here is how I will suggest allocating your funds:

- Rent: 24% of income
- Groceries: 8% of income
- Utility Bills: 4% of income
- Entertainment: 2% of income
- Emergency Fund: 20% of income
- Retirement Savings: 15% of income
- Non-retirement Savings: 10% of income
- Debts: 8.5% of income
- Personal Spending: 8.5% of income



**Figure 2.** Comparative analysis of household savings across three different financial planning scenarios. The 'Original Savings' bars represent the savings calculated from the synthetic dataset without any optimization. The 'Optimized Savings with Bounds' bars illustrate the savings after applying a linear optimization model that maximizes savings while considering minimum essential expenditures. The 'LLM Recommended Savings' bars reflect the aggregate of savings categories as suggested by the LLM, including emergency fund, retirement savings, and non-retirement savings.

In the process of developing our LLM financial recommender system, we mapped the interactions between various components to ensure a seamless user experience and robust performance. Figure 3 illustrates the sequence of events from the moment a user inputs their financial data to the receipt of personalized recommendations and subsequent budget optimization. This diagram delineates the system's workflow, highlighting the role of the LLM in analyzing financial data and generating tailored advice. The sequence further demonstrates the integration of an optimization model that fine-tunes the initial recommendations to accommodate user-specific financial goals and constraints. The interaction between the user, LLM, and optimization model encapsulates the synergy of AI-driven insights with algorithmic precision, which is central to the innovative approach of our financial planning system.



**Figure 3.** Sequence diagram of the LLM financial recommender system illustrating the interaction between the user, LLM system, API interface, optimization model, and database.

The qualitative analysis in Table 1 distinctly showcases the advantages of our proposed models over traditional financial planning methodologies. Key enhancements include the capacity for high personalization, where the LLM's inferential abilities tailor recommendations to specific user scenarios. Unlike traditional models, our approach dynamically adapts to market changes, leveraging the LLM's capacity to analyze current data trends. Scalability is another significant improvement: the AI-driven model can simultaneously cater to a vast array of users, a feat challenging for traditional methods. In terms of user accessibility, the proposed system simplifies complex financial concepts, making financial planning more approachable for the general public. The integration of real-time data ensures that the advice is always relevant and timely, a critical aspect often lacking in conventional models. Finally, the proposed methodology excels in long-term financial planning, with the ability to project and adapt to future financial scenarios, providing users with a comprehensive view of their financial trajectory.

| Criteria                       | Traditional Models | Proposed Models with<br>LLM Integration | Comments  |
|--------------------------------|--------------------|---|---|
| Personalization                | Low                | High                                    | The integration of LLMs allows for highly personalized financial planning based on individual or household-specific data. |
| Adaptability to Market Changes | Limited            | Enhanced                                | LLMs can rapidly adjust recommendations<br>in response to fluctuating market condi-<br>tions.                             |
| Scalability                    | Moderate           | High                                    | The AI-driven approach can easily scale to handle a large number of users simultane-ously.                                |
| User Accessibility             | Variable           | Improved                                | The use of LLMs makes financial planning<br>more accessible to users without deep fi-<br>nancial knowledge.               |
| Real-time Data Integration     | Limited            | Enhanced                                | Proposed models can integrate and analyze real-time data for up-to-date advice.   |
| Long-term Financial Planning   | Moderate           | Superior                                | LLMs can project long-term financial out-<br>comes, aiding in more effective life-cycle<br>financial planning.            |

Table 1. Comparative analysis of traditional and proposed LLM-integrated models.

### 8. Conclusions and Future Work

The transition from traditional optimization approaches to the use of the extended coevolutionary (EC) theory presents an innovative direction for the domain of financial planning. By analyzing cooperative budgeting within households, this paper illuminates the compatibility of game-theoretic models, adaptive learning, and the influential role of large language models (LLMs) in shaping financial recommendations.

The provided pipeline exemplifies practical applications of the EC theory [27], demonstrating a marriage between adaptive learning mechanisms and LLMs to generate financial advisories, as an instantiation of a new economic theory where human decisions can be greatly influenced by AI agents. Through iterative refinements influenced by LLM recommendations, the system iteratively approaches optimal financial allocations. The clear distinction between individualized financial recommendations and cooperative household budgeting reinforces the versatility of this approach, which could be seamlessly integrated into household members by the use of a smartphone, smartwatch, or personal computer. Future work may involve:

- Expansion of Utility Functions: Future studies could delve deeper into constructing more comprehensive utility functions, encompassing factors like changing financial landscapes, inflation rates, and socio-economic considerations.
- Model Robustness: Analyzing the robustness of LLM recommendations in diverse ٠ financial scenarios, especially under market volatility, would be beneficial. This ensures the system's resilience and adaptability to external shocks.
- Integration with Real-time Data: By interfacing the proposed methodology with realtime financial databases, the recommendations can be rendered even more timely and relevant.
- Human-in-the-loop Mechanism: While LLMs are proficient, human expertise remains indispensable. Incorporating a human-in-the-loop mechanism, where financial experts refine or validate LLM recommendations, could elevate the system's reliability.
- Ethical Considerations: As with any AI-driven system, ethical considerations, especially in a domain as crucial as finance, are paramount. Future work must ensure that the recommendations are unbiased, transparent, and in the best interest of the users.

• Scalability to Larger Cooperative Entities: While households were the primary focus, scaling the model to address cooperative budgeting for larger entities, like community groups or small businesses, could be a natural progression.

In essence, the integration of the EC framework into financial planning is but the tip of the iceberg. The potential applications and refinements are vast, opening avenues for research and practical implementations that can significantly enhance the quality of financial advisories and decision-making processes.

Author Contributions: Author contributions: conceptualization, J.d.C. and I.d.Z.; funding acquisition, G.R., C.T.C., and I.d.Z.; investigation, I.d.Z. and J.d.C.; methodology, I.d.Z., J.d.C., and G.R.; software, J.d.C. and I.d.Z.; supervision, C.T.C. and G.R.; writing—original draft, J.d.C. and I.d.Z.; writing—review and editing, J.d.C., I.d.Z., C.T.C., and G.R. All authors have read and agreed to the published version of the manuscript.

**Funding:** We thank the following funding sources from GOETHE-University Frankfurt am Main; 'DePP—Dezentrale Plannung von Platoons im Straßengüterverkehr mit Hilfe einer KI auf Basis einzelner LKW', 'Center for Data Science & AI', and 'xAIBiology'. We acknowledge the support of R&D project PID2021-122580NB-I00, funded by MCIN/AEI/10.13039/501100011033 and ERDF.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data are contained within the article.

**Conflicts of Interest:** The authors declare that they have no conflicts of interest. The funders had no role in the design of the study, in the collection, analyses, or interpretation of data, in the writing of the manuscript, or in the decision to publish the results.

## Abbreviations

The following abbreviations are used in this manuscript:

| Large Language Models          | LLMs |
|--------------------------------|------|
| Deep Learning                  | DL   |
| Artificial Neural Networks     | ANNs |
| Multi-Agent Systems            | MASs |
| Artificial Intelligence        | AI   |
| Extended Coevolutionary        | EC   |
| Retrieval Augmented Generation | RAG  |

## References

- 1. Campbell, J.Y. Household finance. J. Financ. 2006, 61, 1553–1604. [CrossRef]
- Guiso, L.; Sodini, P. Household finance: An emerging field. In *Handbook of the Economics of Finance*; Elsevier: Amsterdam, The Netherlands, 2013; Volume 2; pp. 1397–1532.
- Vaswani, A.; Shazeer, N.; Parmar, N.; Uszkoreit, J.; Jones, L.; Gomez, A.N.; Kaiser, Ł.; Polosukhin, I. Attention is all you need. In Proceedings of the Advances in Neural Information Processing Systems 30, (NIPS 2017), Long Beach, CA, USA, 4–9 December 2017; pp. 5998–6008.
- Radford, A.; Kim, J.W.; Hallacy, C.; Ramesh, A.; Goh, G.; Agarwal, S.; Sastry, G.; Askell, A.; Mishkin, P.; Clark, J.; et al. Learning transferable visual models from natural language supervision. In Proceedings of the 38th International Conference on Machine Learning, Virtual, 18–24 July 2021.
- Perry, V.G.; Martin, K.; Schnare, A. Algorithms for All: Can AI in the Mortgage Market Expand Access to Homeownership? *AI* 2023, 4, 888–903. [CrossRef]
- Roccetti, M.; Delnevo, G.; Casini, L.; Mirri, S. An alternative approach to dimension reduction for pareto distributed data: A case study. J. Big Data 2021, 8, 39. [CrossRef] [PubMed]
- 7. Markowitz, H. Portfolio Selection. J. Financ. 1952, 7, 77–91. [CrossRef]
- Ando, A.; Modigliani, F. The 'Life Cycle' Hypothesis of Saving: Aggregate Implications and Tests. *Am. Econ. Rev.* 1963, 53, 55–84. Available online: http://www.jstor.org/stable/1817129 (accessed on 8 November 2023).
- Charnes, A.; Cooper, W.W. Management Models and Industrial Applications of Linear Programming. *Manag. Sci.* 1957, 4, 38–91. [CrossRef]

- Merton, R.C. Lifetime Portfolio Selection under Uncertainty: The Continuous-Time Case. *Rev. Econ. Stat.* 1969, 51, 247–257. [CrossRef]
- 11. Thaler, R.H. Toward a Positive Theory of Consumer Choice. J. Econ. Behav. Organ. 1980, 1, 39-60. [CrossRef]
- 12. Kahneman, D. Maps of Bounded Rationality: Psychology for Behavioral Economics. *Am. Econ. Rev.* 2003, *93*, 1449–1475. [CrossRef]
- 13. Dixon, M.; Klabjan, D.; Bang, J.H. Classification-based Financial Markets Prediction using Deep Neural Networks. *Algorithmic Financ.* 2017, *6*, 67–77. [CrossRef]
- 14. Mhlanga, D. Industry 4.0 in Finance: The Impact of Artificial Intelligence (AI) on Digital Financial Inclusion. *Int. J. Financ. Stud.* **2020**, *8*, 45. [CrossRef]
- 15. Althnian, A. Design of a Rule-Based Personal Finance Management System Based on Financial Well-Being. *Int. J. Adv. Comput. Sci. Appl.* **2021**, *12*, 182–187. [CrossRef]
- Nazareth, N.; Reddy, Y.Y.R. Financial Applications of Machine Learning: A Literature Review. *Expert Syst. Appl.* 2023, 219, 119640. [CrossRef]
- 17. Dowling, M.; Lucey, B. ChatGPT for (Finance) Research: The Bananarama Conjecture. *Financ. Res. Lett.* **2023**, *53*, 103662. [CrossRef]
- 18. Weber, P.; Carl, K.V.; Hinz, O. Applications of Explainable Artificial Intelligence in Finance—A Systematic Review of Finance, Information Systems, and Computer Science Literature. *Manag. Rev. Q.* **2023**, 1–41. [CrossRef]
- 19. Kumar, S.; Lim, W. M.; Sivarajah, U.; Kaur, J. Artificial Intelligence and Blockchain Integration in Business: Trends from a Bibliometric-Content Analysis. *Inf. Syst. Front.* 2023, 25, 871–896. [CrossRef] [PubMed]
- Pallathadka, H.; Ramirez-Asis, E.H.; Loli-Poma, T.P.; Kaliyaperumal, K.; Ventayen, R.J.M.; Naved, M. Applications of Artificial Intelligence in Business Management, E-commerce and Finance. *Mater. Today Proc.* 2023, 80, 2610–2613. [CrossRef]
- Arsenyan, J.; Piepenbrink, A. Artificial Intelligence Research in Management: A Computational Literature Review. *IEEE Trans. Eng. Manag.* 2023. [CrossRef]
- 22. Xiao, J.J. Applying behavior theories to financial behavior. In *Handbook of Consumer Finance Research;* Springer: New York, NY, USA , 2008; pp. 69–81. .\_5. [CrossRef]
- 23. Browning, M.; Lusardi, A. Household Saving: Micro Theories and Micro Facts. J. Econ. Lit. **1996**, 34, 1797–1855. Available online: http://www.jstor.org/stable/2729595 (accessed on 8 November 2023).
- 24. Branzei, R.; Dimitrov, D.; Tijs, S. *Models in Cooperative Game Theory*; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2008; Volume 556.
- 25. Baker, H.K.; Filbeck, G.; Ricciardi, V. (Eds.). *Financial Behavior: Players, Services, Products, and Markets, Financial Markets and Investments*; Oxford Academic: New York, NY, USA, 2017.
- Arner, D.W.; Zetzsche, D.A.; Buckley, R.P.; Barberis, J.N. FinTech and RegTech: Enabling Innovation While Preserving Financial Stability. *Georget. J. Int. Aff.* 2017, 18, 47–58. Available online: http://www.jstor.org/stable/26395923 (accessed on 8 November 2023). [CrossRef]
- 27. de Zarzà, I.; de Curtò, J.; Roig, G.; Manzoni, P.; Calafate, C.T. Emergent Cooperation and Strategy Adaptation in Multi-Agent Systems: An Extended Coevolutionary Theory with LLMs. *Electronics* **2023**, *12*, 2722. [CrossRef]
- 28. Alhenawi, Y.; Hassan, M.K.; Hasan, R. Evolution of research in finance over the last two decades—A topographical view. *Res. Int. Bus. Financ.* **2022**, *59*, 101550. [CrossRef]
- 29. Luigi, G.; Luana, Z. From patriarchy to partnership: Gender equality and household finance. J. Financ. Econ. 2023, 147, 573–595.
- 30. Chai, J.; Horneff, W.; Maurer, R.; Mitchell, O.S. Optimal Portfolio Choice over the Life Cycle with Flexible Work, Endogenous Retirement, and Lifetime Payouts. *Rev. Financ.* 2011, *15*, 875–907. [CrossRef]
- 31. Browning, M.; Crossley, T.F. The Life-Cycle Model of Consumption and Saving. J. Econ. Perspect. 2001, 15, 3–22. . [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.